# Antidepressant treatment with sertraline for adults with depressive symptoms in primary care: the PANDA research programme including RCT

Larisa Duffy,<sup>1\*</sup> Gemma Lewis,<sup>1</sup> Anthony Ades,<sup>2</sup> Ricardo Araya,<sup>3</sup> Jessica Bone,<sup>1</sup> Sally Brabyn,<sup>4</sup> Katherine Button,<sup>5</sup> Rachel Churchill,<sup>6</sup> Tim Croudace,<sup>7</sup> Catherine Derrick,<sup>2</sup> Padraig Dixon,<sup>2</sup> Christopher Dowrick,<sup>8</sup> Christopher Fawsitt,<sup>2</sup> Louise Fusco,<sup>8</sup> Simon Gilbody,<sup>4</sup> Catherine Harmer,<sup>9</sup> Catherine Hobbs,<sup>5</sup> William Hollingworth,<sup>2</sup> Vivien Jones,<sup>2</sup> Tony Kendrick,<sup>10</sup> David Kessler,<sup>2</sup> Naila Khan,<sup>8</sup> Daphne Kounali,<sup>2</sup> Paul Lanham,<sup>11</sup> Alice Malpass,<sup>2</sup> Marcus Munafo,<sup>12</sup> Jodi Pervin,<sup>4</sup> Tim Peters,<sup>2</sup> Derek Riozzie,<sup>11</sup> Jude Robinson,<sup>13</sup> George Salaminios,<sup>1</sup> Debbie Sharp,<sup>2</sup> Howard Thom,<sup>2</sup> Laura Thomas,<sup>2</sup> Nicky Welton,<sup>2</sup> Nicola Wiles,<sup>2</sup> Rebecca Woodhouse<sup>4</sup> and Glyn Lewis<sup>1</sup>

- <sup>1</sup>Division of Psychiatry, University College London, London, UK
- <sup>2</sup>Bristol Medical School, University of Bristol, Bristol, UK
- <sup>3</sup>Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK
- <sup>4</sup>Department of Health Sciences, University of York, York, UK
- <sup>5</sup>Department of Psychology, University of Bath, Bath, UK
- <sup>6</sup>Centre for Reviews and Dissemination, University of York, York, UK
- <sup>7</sup>School of Nursing and Health Studies, University of Dundee, Dundee, UK <sup>8</sup>Institute of Psychology Health and Society, University of Liverpool, Liverpool, UK
- <sup>9</sup>Department of Psychiatry, University of Oxford, Oxford, UK
- <sup>10</sup>Primary Care and Population Sciences, Faculty of Medicine, University of Southampton, Southampton, UK
- <sup>11</sup>Patient and public involvement contributor, UK
- <sup>12</sup>Department of Psychology and Integrated Epidemiology Unit, University of Bristol, Bristol, UK
- <sup>13</sup>Department of Sociology, Social Policy and Criminology, University of Liverpool, Liverpool, UK

\*Corresponding author larisa.duffy@ucl.ac.uk

Declared competing interests of authors: Rachel Churchill reports grants from the National Institute for Health Research (NIHR) (Programme Grants for Applied Research programme) during the conduct of the study. Simon Gilbody serves as deputy chairperson of the NIHR Health Technology Commissioning Board, but was not involved in the commissioning of this programme of research. Catherine Harmer reports personal fees from P1vital (Wallingford, UK), grants from UCB Pharma (Brussels, Belgium), grants and personal fees from Johnson & Johnson (New Brunswick, NJ, USA), and personal fees from H. Lundbeck A/S (Copenhagen, Denmark), Servier Laboratories (Neuilly-sur-Seine, France) and Pfizer Inc. (New York, NY, USA) outside the submitted work. Tony Kendrick reports grants from NIHR during the conduct of the study. Marcus Munafo reports grants and personal fees from Cambridge Cognition (Cambridge, UK) and personal fees from Jericoe Ltd (Bristol, UK) outside the submitted work. Tim Peters reports grants from NIHR during the conduct of the study. Howard Thom reports personal fees from Novartis Pharma AG (Basel, Switzerland), Pfizer Inc., Roche Holding AG (Basel, Switzerland) and Eli Lilly and Company (Indianapolis, IN, USA) outside the submitted work. Nicky Welton reports grants from NIHR during the conduct of the study; and she is the principal investigator on a Medical Research Council-funded project in collaboration with Pfizer Inc. Pfizer Inc. part funded a junior researcher on the project. The project is purely methodological using historical data on pain relief, and unrelated to this work. Nicola Wiles reports grants from NIHR during the conduct of the study. Glyn Lewis reports grants from University College London during the conduct of the study and personal fees from Fortitude Law (London, UK) outside the submitted work.

Published December 2019 DOI: 10.3310/pgfar07100

## **Plain English summary**

## The PANDA research programme including RCT

Programme Grants for Applied Research 2019; Vol. 7: No. 10 DOI: 10.3310/pgfar07100

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

There were over 70 million antidepressent prescriptions in England in 2018, and there are concerns that they are overprescribed. The aim of the PANDA (What are the indications for Prescribing ANtiDepressAnts that will lead to a clinical benefit?) research programme was to provide general practitioners with improved guidance to help them make recommendations about the likely response to antidepressants.

We carried out two studies. In the first, we recruited 558 people with depressive symptoms and followed them for 6 weeks. We estimated a patient-centred measure of the minimal clinically important difference, an improvement that would be recognised by the patient. We found that the minimal clinically important difference was best expressed as about a 20% reduction in the patient's initial symptoms, but at lower symptoms the percentage change for the minimal clinically important difference became larger.

We also interviewed patients with open-ended interviews. They reported that the Patient Health Questionnaire-9 items failed to fully capture their experience of recovery from depression. Some patients struggled with how the questions were phrased and there was a lot of disagreement between self-reported improvement and change in Patient Health Questionnaire-9 items score. We concluded that clinicians should not rely on scales such as the Patient Health Questionnaire-9 items to assess improvement but should use additional questions and further clinical assessment before deciding on improvement.

Our second study was a randomised clinical trial (on 653 participants) to investigate the clinical effectiveness of sertraline, a commonly used first-line antidepressant. We found no evidence that sertraline was more effective than a placebo (an identical inactive capsule) at reducing depression symptoms by 6 weeks, but by 12 weeks we found some evidence for a reduction in depressive symptoms. However, we found strong evidence that sertraline was effective at reducing anxiety symptoms and that patients who took sertraline were more likely to report improvement and better mental health overall. Sertraline is a cheap intervention that had a high probability of being cost-effective. We found no evidence that the severity and duration of patients' depressive symptoms predicted their response to antidepressants or the cost-effectiveness of treatment.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Duffy et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

## **Programme Grants for Applied Research**

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/ programme-grants-for-applied-research.htm

#### This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0610-10048. The contractual start date was in March 2012. The final report began editorial review in September 2018 and was accepted for publication in June 2019. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Duffy *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### **NIHR Journals Library Editor-in-Chief**

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk